Author's response to reviews

Title: Comparative Effectiveness of Asthma Interventions within a Practice Based Research Network

Authors:

Hazel Tapp (hazel.tapp@carolinashealthcare.org)
Lisa Hebert (lisa.hebert@carolinashealthcare.org)
Michael Dulin (Michael.dulin@carolinashealthcare.org)

Version: 4 Date: 15 August 2011

Author's response to reviews:

Following changes made:

Author list: 'Michael Dulin' in the submission system, 'Michael Dulin' in the manuscript.

Figures: provided titles for all figures.
Minor revisions

Affiliations: included the country in the affiliation details.

Thank you very much. Sincerely, Hazel Tapp